Author’s response to reviews

Title: Perceived stigma and barriers to care in UK Armed Forces personnel and veterans with and without probable mental disorders

Authors:

Victoria Williamson (victoria.williamson@kcl.ac.uk)

Neil Greenberg (neil.greenberg@kcl.ac.uk)

Sharon Stevelink (sharon.stevelink@kcl.ac.uk)

Version: 4 Date: 01 Nov 2019

Author’s response to reviews:

-- In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

We have listed that the funder did not influence the design/data collection or analysis/writing of the paper.

-- If abbreviations are used in the text they should be defined in the text at first use, and a list of abbreviations should be provided in the Declarations.

We have listed these abbreviations in the declarations section.

--Please ensure that 'Neil Greenberg' in the author list has been associated to an institution via numbering/lettering in superscript.

We have added Prof Greenberg's affiliation.

-- At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

We have attached a copy that has no tracked changes.